Author's response to reviews

Title: Facilitators and barriers to healthy eating practices and physical activity among adolescent girls in rural South Africa

Authors:

Modiehi H Sedibe (sedibemh@gmail.com)
Kathleen Kahn (Kathleen.Kahn@wits.ac.za)
Kerstin Edin (kerstin.edin@umu.se)
Tabby Gitau (tabbymuthoni@gmail.com)
Anneli Ivarsson (anneli.ivarsson@epiph.umu.se)
Shane A Norris (Shane.Norris@wits.ac.za)

Version: 2 Date: 7 February 2014

Author's response to reviews:

Nutrition journal recommended that we submit our manuscript to BMC Pediatrics journal